

## G OPEN ACCESS

**Citation:** Fox SJ, Bellan SE, Perkins TA, Johansson MA, Meyers LA (2019) Downgrading disease transmission risk estimates using terminal importations. PLoS Negl Trop Dis 13(6): e0007395. https://doi.org/10.1371/journal.pntd.0007395

**Editor:** Thomas S. Churcher, Imperial College London, UNITED KINGDOM

Received: July 13, 2018

Accepted: April 16, 2019

Published: June 14, 2019

**Copyright:** This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative</u> Commons CC0 public domain dedication.

**Data Availability Statement:** All files to replicate the methodology contained within the paper can be found here: doi:10.18738/T8/HYZ53B. We were not given permission to share the true Texas county importation time series data, but have provided a fake dataset to evaluate the method.

**Funding:** This publication was supported by Cooperative Agreement Number U01CK000512, funded by the Centers for Disease Control and Prevention. Its contents are solely the responsibility of the authors and do not necessarily **RESEARCH ARTICLE** 

# Downgrading disease transmission risk estimates using terminal importations

# Spencer J. Fox<sup>1\*</sup>, Steven E. Bellan<sup>2,3</sup>, T. Alex Perkins<sup>4</sup>, Michael A. Johansson<sup>5,6</sup>, Lauren Ancel Meyers<sup>1,7</sup>

Department of Integrative Biology, University of Texas at Austin, Austin, Texas, United States of America,
 Department of Epidemiology and Biostatistics, University of Georgia, Athens, Georgia, United States of America,
 Center for Ecology of Infectious Diseases, University of Georgia, Athens, Gerogia, United States of America,
 Department of Biological Sciences and Eck Institute for Global Health, University of Notre Dame, Notre Dame, Indiana, United States of America,
 Division of Vector-Borne Diseases, Centers for Disease Control and Prevention, San Juan, Puerto Rico,
 Center for Communicable Disease Dynamics, Harvard TH Chan School of Public Health, Boston, Massachusetts, United States of America,
 Santa Fe, New Mexico, United States of America

\* spncrfx@gmail.com

### Abstract

As emerging and re-emerging infectious arboviruses like dengue, chikungunya, and Zika threaten new populations worldwide, officials scramble to assess local severity and transmissibility, with little to no epidemiological history to draw upon. Indirect estimates of risk from vector habitat suitability maps are prone to great uncertainty, while direct estimates from epidemiological data are only possible after cases accumulate and, given environmental constraints on arbovirus transmission, cannot be widely generalized beyond the focal region. Combining these complementary methods, we use disease importation and transmission data to improve the accuracy and precision of a priori ecological risk estimates. We demonstrate this approach by estimating the spatiotemporal risks of Zika virus transmission throughout Texas, a high-risk region in the southern United States. Our estimates are, on average, 80% lower than published ecological estimates—with only six of 254 Texas counties deemed capable of sustaining a Zika epidemic—and they are consistent with the number of autochthonous cases detected in 2017. Importantly our method provides a framework for model comparison, as our mechanistic understanding of arbovirus transmission continues to improve. Real-time updating of prior risk estimates as importations and outbreaks arise can thereby provide critical, early insight into local transmission risks as emerging arboviruses expand their global reach.

#### Author summary

When novel infectious diseases like chikungunya or Zika emerge and threaten global spread, public health officials worldwide must assess the risk for local introductions and outbreaks. These assessments are made in anticipation of local case data, and officials must draw upon historic evidence from similar diseases or locations. Thus, accurate local risk assessments have most often been limited to retrospective analyses and have been



represent the official views of the Centers for Disease Control and Prevention or the Department of Health and Human Services. SJF and LAM would like to acknowledge funding from the Models of Infectious Disease Agent Study (MIDAS) program grant number U01GM087719; SEB received support from a National Institute of Health National Institute of Allergy and Infectious Diseases grant (K01AI125830); MAJ received partial support from the MIDAS program (Cooperative Agreement number 1U54GM088558); and TAP received support from the National Science Foundation (DEB 1641130) and the Defense Advanced Research Projects Agency (D16AP00114). The sponsors had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

unavailable in real-time during an emerging epidemic. Here we present a method that can harness both historic and current data to produce early risk assessments and update projections in real-time. We demonstrate our approach by estimating local transmission risk for Zika throughout the state of Texas. Our findings suggest that the majority of Texas counties face little risk for sustaining a Zika epidemic, and successfully predict the number of locally transmitted cases across the state. Real-time updating of local transmission risk estimates during an emerging epidemic can thus provide actionable, early insight for public health response as emerging arboviruses expand their global reach.

#### Introduction

The explosive emergence of Zika virus (ZIKV) in the Americas in 2016 caught the global health community by surprise. Officials scrambled not only to control the disease at its source but also to anticipate and rapidly contain global transmission via infected travelers [1]. The rate at which a newly introduced pathogen spreads can vary enormously though, particularly for ZIKV and other pathogens with multi-faceted drivers of their transmission [2]. For example, serological surveys of human exposure to the ecologically similar dengue virus (DENV) on either side of the Texas-Mexico border indicated far higher DENV exposure in the Mexican community, despite virtually identical climatic conditions and even higher mosquito abundance in the Texan community [3]. This result suggests that *a priori* prediction of risk may be quite challenging.

Epidemiological risk assessment—estimating the severity and transmissibility of a threatening disease—can be vital to successful mitigation with limited resources. Historical outbreak data can provide invaluable insight into future epidemic risk. However, for a pathogen that has yet to arrive or has just begun to spread, we are forced to borrow epidemiological data from other populations or related pathogens, or to indirectly assess risk based on environmental suitability. For example, as the first importations of ZIKV arrived in the US in 2016, early attempts to determine the likelihood and rate of local transmission relied primarily on dengue epidemiological data from regions with markedly different climatic and socioeconomic conditions [4–6].

These ecological risk assessments provide information regarding the basic reproduction number of a disease ( $R_0$ )—the expected number of secondary human infections resulting from a single human infection—which provides a meaningful and predictive measure of local epidemiological risk. For ZIKV,  $R_0$  encompasses the combined effects of the multi-step transmission life cycle: a mosquito biting an infected individual, incubating the disease and becoming infected, and finally biting and infecting a new susceptible individual. In a naive population,  $R_0$  indicates whether importations can potentially lead to sustained local epidemics; if so, it also provides insight into the probability, magnitude, and speed of spread [7, 8]. However, ecological estimates often carry considerable uncertainty stemming from model parameterization and regional extrapolation, and suggest a wide range of possible epidemic outcomes, from terminal importations to stuttering chains of transmission to full blown epidemics [4, 9–11]. Once an outbreak is underway, early case data can be used to directly estimate  $R_0$  [12–14]. For arboviruses with environmentally constrained spatial heterogeneity, such case-based estimates cannot be easily extrapolated from one region to others.

Here, we introduce a method for performing real-time updating of ecologically informed estimates of  $R_0$  that reconciles discrepancies between initial estimates of  $R_0$  and real-time data on local transmission of an emerging arbovirus. This approach is applicable to settings in

which there is an ongoing threat of arbovirus importation and in which transmission has not precipitated a large-scale epidemic. In particular, this work was motivated by recent introductions of ZIKV into the continental US. As hundreds of cases arrived from affected regions throughout the Americas, officials sought to estimate risks of autochthonous (local) transmission and identify high-risk regions in the southern US. However, given the novelty of ZIKV and the large proportion of ZIKV cases that go undetected, early ecological estimates had high uncertainty [5, 6, 15–17]. Our method combines indirect and direct estimation methods to reduce such uncertainty and increase accuracy at high spatiotemporal resolution. We first build prior ecological estimates of local  $R_0$  and then harness real-time importation data—cases that arrive in a naive location with or without subsequently infecting others—to update the estimates, while explicitly modeling case reporting uncertainty. As a case study, we use the almost complete absence of secondary transmission following 298 importations of ZIKV into the state of Texas in 2016 and 2017 to lower and narrow local estimates of  $R_0$ .

#### Materials and methods

We used a two-step procedure to estimate the monthly  $R_0$  for each of the 254 Texas counties (hereafter county-month  $R_0$ ): (1) estimate *a priori* county-month  $R_0$  distributions using published ecological models of ZIKV transmission [4, 6], and (2) using these as Bayesian priors, generate posterior  $R_0$  distributions based on reported importations and subsequent local transmission.

#### Data

We analyzed all known ZIKV importations into Texas from January 2016 to September 2017, including the county and notification date; county-level purchasing power parity (PPP) in US dollars [18]; daily temperature data at a 5 km x 5 km resolution for 2016-2017 and historical averages from 1960-1990 [19, 20]. For each county and month, we averaged daily temperatures across all 5 km x 5 km grid cells whose center fell within the county; we aggregated 5 km x 5 km mosquito (*Aedes aegypti*) occurrence probabilities similarly [21]. Those data are available at dx.doi.org/10.18738/T8/HYZ53B.

In all, six mosquito-borne, autochthonous cases of ZIKV were reported in Texas in 2016 and one was reported in 2017 [22]. For updating  $R_0$  estimates, we analyzed 2016 data where two autochthonous cases were detected in Cameron County from passive surveillance–one in November and one in December 2016; we excluded four nearby cases discovered during the November follow-up investigation, because our model does not incorporate active surveillance. As sensitivity analyses, we re-estimated  $R_0$  assuming that no cases were detected and that all six cases were detected (S1 Fig). Accounting for this worst-case scenario lead to increased estimates for risk across the state, but expected risk still remained well below epidemic potential. For validating our estimates, we analyzed 2017 data and considered only one of the two reported autochthonous cases, as the second case occurred outside the timeline of our 2017 importation data.

#### A priori county-month R<sub>0</sub> estimates

For ZIKV to be locally transmitted in a location, a mosquito must bite an individual, the mosquito must be infectious with the virus, and then that mosquito must bite a susceptible individual and transmit that virus. We focus our analysis on a singular value,  $R_0$ , which captures the individual contributions of these factors on the potential for human to human transmission. Following Perkins *et al.* [6], we estimated  $R_0$  using a temperature-dependent Ross-Macdonald formulation,

$$R_0(T) = \frac{mbca^2 e^{-\mu(T)n(T)}}{\mu(T)r},$$
(1)

with parameters as defined in Table 1. Although others have considered temperature dependence in parameters other than  $\mu$  and n (e.g., [23, 24]), we opted for our relatively simple formulation for a few reasons. First, effects of temperature on  $\mu$  and n are likely to be the most critical to include, due to their relatively strong influence on  $R_0$  [25]. Second, whereas temperature can influence m in multiple ways [23], we elected to model m in an alternative fashion that incorporates influences from a wider range of factors than temperature alone. Third, our approach for scaling  $R_0$  through data assimilation—which is the primary goal of this work—is applicable to essentially any formulation of *a priori* county-month  $R_0$ . The  $R_0$  prior used here is particularly germane to the real-time nature of this exercise, given that an early (but largely similar) version of it was available as a preprint near the very beginning of the time window of our analysis [26].

The ratio of mosquitoes to people, *m*, makes use of two spatially varying inputs: *Ae. aegypti* occurrence probability and an index of economic purchasing power. The first of these was based on a global collection of *Ae. aegypti* occurrence records used to inform an occurrence probability model that made use of a number of environmental variables, including a temperature suitability index, precipitation, enhanced vegetation index, and urbanicity [21]. The second of these features spatial estimates of purchasing power parity (PPP) [18], which offers a relative spatial index of economic wealth that other work [6, 32] has shown can serve as a useful proxy for spatial variation in economic factors affecting mosquito-human contact. Here, we used gridded estimates of those two spatially variable inputs for Texas, but otherwise followed the methodology of Perkins *et al.* [6] for translating them into a gridded surface of *m*. In brief, this involved transforming mosquito occurrence probability into a proxy for mosquito abundance and multiplying it by a shape-constrained additive model of PPP, which resulted in a convex, monotonically decreasing relationship between PPP and *m*. We do not consider the near-term effects of precipitation on local ZIKV transmission risk, but one could be included in future iterations, given a validated *a priori* model describing the relationship.

To derive *a priori* distributions of  $R_0$  for each county-month, we drew 1,000 Monte Carlo samples from each parameter underlying the  $R_0$  formulation described above. This accounted for uncertainty in  $\mu$ , *n*, *Ae. aegypti* occurrence probability, and the relationship between PPP and *m*, consistent with previous descriptions of uncertainty for each of those parameters [6]. These parameter draws were applied to the appropriate county and month data for Texas. Finally, we fit gamma distributions to each probability distribution for local transmission risk for use as informative priors, see S2 Fig for comparisons between the samples and the fitted gamma distributions.

#### Autochthonous transmission likelihood

We developed a likelihood function describing the expected outbreak size following an importation, using the approach from [13]. Assuming that the secondary case distribution for each infected individual is negative binomial with mean  $R_0$  and dispersion parameter, k, and that all cases are detected, then the probability of an outbreak of chain size, j, is can be described by [13, 14] as

$$s(j, R_0, k) = \frac{\Gamma(kj+j-1)}{\Gamma(kj)\Gamma(j+1)} \frac{(R_0/k)^{j-1}}{(1+(R_0/k))^{kj+j-1}}$$
(2)

#### Table 1. Parameters used to create prior $R_0$ estimates from the literature.

Parameter	Description	Distribution	Value (CI)	Source
b	Mosquito-to-human transmission probability	Constant	0.4	[27]
c/r	Human-to-mosquito transmission probability times the duration of human infectiousness (days)	Constant	3.5	[28]
а	Mosquito biting rate	Constant	0.67	[29]
$\mu(T)$	Mosquito daily mortality rate	Generalized additive model	$0.00002 (0.000008, 0.287)^1$	[25, 30]
n(T)	Extrinsic incubation period in mosquitoes (days)	Exponential	6.1 (3.4, 9.9) <sup>1</sup>	[31]
т	Mosquito-human ratio	Spatially variable with <i>Ae. aegypti</i> and G-Econ indices	0.36 (0.20, 0.60)	[6]

#### <sup>1</sup>. At 30 °C

https://doi.org/10.1371/journal.pntd.0007395.t001

where  $\Gamma(n) = (n - 1)!$ . However, not all cases are detected and we assume the imported index case is always detected and correctly classified as an importation, so the probability of detecting a chain of size, *j*, from a given importation is given by

$$s^{*}(j, R_{0}, k, p_{d}) = \sum_{l=j}^{\infty} s(l, R_{0}, k) \cdot \binom{l-1}{j-1} \cdot p_{d}^{j-1} \cdot (1-p_{d})^{l-j}$$
(3)

where  $p_d$  is the case detection probability [13, 14]. This equation can be understood intuitively. If cases can be missed then there are infinitely many ways in which an outbreak of a certain size can be detected (e.g. 5 detected cases could stem from an outbreak of size 6 with 1 case missed, or size 7 with 2 missed etc). This equation enumerates those possibilities and sums their individual marginal probabilities. Importantly, this allows for local, undetected cases. We take the product of all likelihoods for each imported case as

$$\mathcal{L}(\vec{O}|\alpha, \vec{R_0}, k, p_d) = \prod_{i=1}^{\text{length}(\vec{O})} s^*(O_i, \alpha R_{0\gamma_i, \omega_i}, k, p_d)$$
(4)

where  $\vec{O}$ , contains the observed outbreak sizes for each importation (terminal importations have an outbreak size of one),  $R_{0\gamma_i,\omega_i}$  denotes the county( $\gamma$ )-month( $\omega$ )  $R_0$  for the location and time that the importation occurred, and  $\alpha$  is a statewide scaling factor applied to each  $R_{0\gamma_i,\omega_i}$ . The introduction of the state-wide scaling factor allows for localized importations to inform statewide estimates, but assumes that biases in the *a priori*  $R_0$  estimation procedure are constant across counties and months. While our analysis does not assume that all imported cases are detected, we do not explicitly model missed imported cases and simply explore the consequences of these potentially missed cases in a sensitivity analysis. If systematic biases in the spatiotemporal detection of imported cases are discovered, our method should be updated accordingly. Details of simulations and validation of the likelihood can be found in S1 File and S3 Fig.

#### Estimating the dispersion parameter

The negative binomial dispersion parameter governs the variability in secondary cases following each importation, with values near zero meaning that most importations yield few or no cases while a few "superspreaders" produce many. Superspreading dynamics are known to occur with mosquito-borne diseases like ZIKV [33, 34]. We assume that ZIKV secondary case distributions are the same as that of dengue virus (DENV) [35], though these estimates should be updated as ZIKV-specific secondary case distributions are identified. Padmanabha et al. describe the relationship between regional  $R_0$  and the percentage of DENV imported cases that generate over 20 secondary infections ( $p_{20}$ ), as  $R_0 = 0.63 \times 100 \times p_{20} + 0.58$ . As we require  $p_{20} > 0$  for all values of  $R_0$ , we set  $p_{20}$  to an arbitrarily low, but non-zero value ( $p_{20} = 1 \times 10^{-8}$ ) for values of  $R_0 < 0.58$ . We then found the dispersion parameter for every value of  $R_0$  that lead to the expected  $p_{20}$  value from that previously estimated relationship. The result was that a single dispersion parameter captured the relationship well for all  $R_0$  values and thus we set the dispersion parameter, k = 0.12, for all analyses (S4 Fig). The highly dispersed nature of the secondary case distribution suggests that most importations will not lead to epidemics, even if  $R_0$ > 1. For example, even with  $R_0 = 1.5$ , we would only expect to see an epidemic stemming from ~8% of imported cases.

#### Updating posterior R<sub>0</sub> estimates

Given our set of prior distributions for county-month transmission risk, we set out to estimate the statewide scaling factor,  $\alpha$ , as a means to update our *a priori* transmission risk estimates. We estimate the posterior distribution for this scaling factor for each day with a new importation between January 2016 and January 2017 (our method also explicitly updates countymonth  $R_0$  estimates for county-month combinations with imported cases). Importantly, this doesn't indicate that  $\alpha$  is changing through time, but rather that we refine our estimate for it as more data accumulate. We assumed a uniform prior for  $\alpha$  of  $U \sim (0, 2)$ , and used a blocked Gibbs sampling algorithm of MCMC. For each MCMC step we provide the detected imported cases to date and propose each county-month  $R_0$ , a single  $\alpha$ , and a probability of case detection,  $p_d$ . County-month  $R_0$  proposals were normally distributed around the previous sample with standard deviation of 0.1,  $\alpha$  proposals were distributed  $U \sim (0, 2)$ . We used a previously estimated distribution for  $p_d$  as a strong, informative prior,  $p_d \sim N(5.74\%, sd = 1.49\%)$ , as there are identifiability issues given its relationship with  $\alpha$ , and assumed it to not vary spatiotemporally [32]. There were no differences between the posterior and prior distributions for the reporting rate S5 Fig. We used the Metropolis-Hastings probability to accept or reject proposals. Our chains consisted of 200,000 samples with a burn-in duration of 100,000, thinning every 10 steps. We ran a single chain for each parameter set, and assessed convergence through visual inspection of the trace plots (S6 Fig). Further algorithmic details and code are available on Github (https://github.com/sjfox/rnot\_updater). To help with an intuitive understanding of the method we also present a hypothetical situation (S1 File and S7 Fig).

#### Validating posterior county-month R<sub>0</sub> estimates

We derived the expected number of autochthonous cases from the importation data through September 2017 (at that time, the most recent importation was detected in mid-May) and compared the estimates to the actual reported autochthonous cases. We incorporated uncertainty into our estimates by sampling from the posterior county-month  $R_0$  distributions and simulating outbreaks accordingly (full details in S1 File).

#### Results

In the face of a novel emerging infectious disease threat, public health officials must borrow information from related diseases and/or from epidemics in geographically distinct regions. Adapting these models to new regions necessarily increases the uncertainty in risk estimates, but can prove invaluable for data-driven public health decision-making. Here we present a rational framework for updating these initially uncertain estimates with real-time case data



**Fig 1. Texas importations and baseline transmission risk estimates for 2016-17.** (A) Initial ZIKV  $R_0$  estimates using ecological risk models parameterized with actual 2016-2017 temperatures. Each solid line shows median values for one of Texas' 254 counties. Dashed line shows the highest upper bound (99th percentile) across all counties. Horizontal dotted red line illustrates the threshold for county-month epidemic risk ( $R_0 = 1$ ). (B) Daily ZIKV importations into Texas. Blue arrows indicate importations that produced detected autochthonous transmission; shading indicates training (2016) and testing (2017) periods.

from the emergence of Zika in Texas. The results are structured as such: we first present data from Texas' baseline importation and transmission risk estimates, we then update those estimates with importation and local transmission data, and finally present findings from validating those estimates using out of sample data.

#### Baseline importation and transmission risks in Texas

Prior to making importation-based updates, our initial median estimates of R<sub>0</sub> across Texas' 254 counties in 2016 range from approximately 0 to 1.5 throughout the year with July and August having the highest transmission risk (Fig 1A). Given the potential implications for determining the epidemic potential of a county, throughout the manuscript, we conduct a one-sided test at a 1% significance level of the resultant probability distributions. We thus take a conservative approach and only consider counties whose probability distribution's 99th percentile (upper bound) is above one to be at risk for an epidemic ( $R_0 > 1$ ). Simply put, we only classify counties as no epidemic risk if there is less than a 1% chance of  $R_0 > 1$ . Initial upper bound estimates reach as high as three, and 119 (47%) of Texas counties are expected to be at risk of a local epidemic in at least one month of the year (Fig 1A, S8 Fig). When we considered historic average temperatures rather than 2016 temperatures, the projected 2017 risks were consistently lower, with the largest differences occurring during the unseasonably warm 2017 winter (S9 Fig). Case importations peaked in July, August, and September of 2016, with 164 (55%) of the 298 total 2016 importations arriving then (Fig 1B). The few detected autochthonous cases occurred in November and December, when expected risk was relatively low but not negligible.



Fig 2. Posterior median county  $R_0$  estimates for Texas, based on ZIKV importations through January 2017. This assumes that all importations were terminal except for two autochthonous cases detected in Cameron County in late 2016.

#### Updated transmission risks in Texas

Based on all importations and autochthonous cases that occurred in Texas prior to January 2017, we estimate that all Texas counties have a median posterior  $R_0$  below 1 (Fig 2). Median estimates range from 0 to 0.29; upper-bound estimates range from 0 to 1.12, with only six (5%) of the original 119 high-risk counties maintaining epidemic potential (S10 Fig). The six counties with remaining epidemic risk are all contained within the greater Houston metropolitan area, and all have relatively low estimated transmission risk estimates ranging from 0.25-0.29 (Grimes, Houston, Madison, Montgomery, Walker, and Waller counties). When we assume historic averages rather than 2016 temperatures, we obtain similar results (S11 Fig).

In a sensitivity analysis that assumes  $\sim 20$  times more undetected importations, we found that the estimated risks decreased further (S1 Fig). We also varied the number of detected autochthonous cases in November: as they decrease from one to zero, the estimated risks decrease considerably; as they increase to five, estimated risks increase, with 83 counties becoming at risk for a local outbreak (S1 Fig). However expected transmission risk for these counties is still well below 1, ranging from 0.3-0.6.

Importation events had variable impacts on the posterior estimates, depending on their timing and location (Fig 3). Terminal importations early in the year, when *a priori*  $R_0$  estimates were low, had little effect; those arriving in the summer months, when high *a priori*  $R_0$  estimates suggested that transmission should have occurred, led to sharp decreases and a shrinking confidence interval. By early November, the median  $\alpha$  decreased from 1.0 to 0.06 with a 95% CI of 0.002-0.30. The two secondary transmission events detected in November and



Fig 3. Evolving posterior distribution of statewide scaling factor for  $R_0$ . Zika importations, both with and without subsequent detected autochthonous transmission, provide insight into local transmission potential, via a statewide scaling factor,  $\alpha$ . This shows the posterior distributions of  $\alpha$ , for each day of 2016 that had at least one imported case. Median estimates reach a minimum in early November, just before the detected autochthonous transmission events (upside-down blue triangles). Red shading indicates the average statewide monthly temperature. Note: the scaling factor is never less than zero.

December increased the posterior  $R_0$  estimates and widen the confidence interval slightly, but did not provide enough evidence to qualitatively change the transmission risk estimates. Incorporating all data up to January 2017, our best estimate is that  $R_0$  values across the state are roughly one fifth the original estimates (median: 0.19, 95% CI: 0.05-0.48).

Interestingly, based on the *a priori*  $R_0$  estimates, we would have predicted local transmission to occur during the summer when importations and transmission risk were high. In fact, these initial expectations for transmission were lowered even further prior to the detected locally acquired case in November. Since we estimated the posterior distributions for transmission risk and the statewide scaling factor jointly, we were able to compare the prior and posterior estimates to understand if specific counties were downgraded more or less than expected by the statewide average. We find that the ratio between prior and posterior estimates tends to average near the statewide scaling factor of 0.19, except for counties that experienced many reported importations during the year (S12 Fig). These counties tended to have lower than average ratios like Harris, Dallas, Tarrant, Bexar, and Cameron counties (S12 Fig). While the timing of the autochthonous transmission may not have been predicted using our initial risk estimates, the transmission events both occurred in the southern tip of Texas (Cameron county), which was predicted to be the highest risk county during those months.

The temporal mismatch between the model predictions and the observed ZIKV local transmission in November and December might stem from the a priori transmission risk model, which constrains posterior estimates. Specifically, the model does not account for potential time lags between ZIKV environmental suitability and eventual emergence of locally acquired cases. Environmental determinants of mosquito population growth may take several weeks or months to impact arbovirus risk. Furthermore, the timing of case reports following infection will depend on the timing and severity of symptoms, healthcare seeking behavior and clinical reporting. To assess how this effect might alter results, we incorporated a two month lag in our a priori suitability model as estimated in [36]. In essence we shifted transmission risk estimates two months later than our baseline predictions suggested. For example importations occurring in November are matched with baseline transmission risk estimates from September. The resulting model yielded estimates that are more consistent with the observed ZIKV local transmission, although posterior risk estimates were remarkably consistent with our baseline results, with the highest risk period shifted by two months (S13 and S14 Figs). In comparing our baseline model with the lagged one, we calculate a BIC of 2.9 in favor of the lagged model, suggesting a moderate improvement to model fit.

#### Expected autochthonous transmission in Texas

We use transmission risk estimates based on importations through December 2016 to estimate the number of autochthonous cases we would expect to detect in Texas in 2017. Based on our posterior estimates we estimate a 1% probability of at least one single detected case in Hidalgo county at the time the secondary transmission occurred. Since the sample size is low, we average the expected number of secondary cases over the 2017 time period where importation data were available. Assuming first that only the reported importations occurred in 2017 (26 total), we estimate that there should have been 0.08 (95%CI: 0-1) detected autochthonous cases in the state; assuming that many importations went undetected, according to the reporting probability ( $\frac{26}{p_d} \approx 453$  cases), we estimate 1.3 (95% CI: 0-7) detected autochthonous cases. Both of these estimates are consistent with the single autochthonous case detected in Texas in 2017, though our results best fit a scenario with many undetected importations (Fig 4).

#### Discussion

The global expansion of ZIKV was declared a Public Health Emergency of International Concern in February 2016, and caused more than 565,000 confirmed or probable cases and over 3,352 documented cases of congenital Zika syndrome. Although it is receding in most regions of the world, ecological risk assessments suggest that previously unaffected or minimally



**Fig 4. Expected autochthonous cases in 2017, assuming revised county**  $R_0$  **estimates through September 2017.** The probability distributions are built from 10,000 simulations, each randomly drawing from the  $R_0$  posterior distributions. The dashed blue line indicates the actual number of detected autochthonous cases in state (one), and the solid black lines indicates the mean expected number of autochthonous cases for the baseline importation scenario, in which only the reported importations occurred (top) and the increased importation scenario, in which a large fraction of importations went undetected (bottom).

affected areas may remain at risk for future emergence, including parts of Asia and South America [37-39]. Differentiating regions that can sustain a ZIKV epidemic ( $R_0 > 1$ ) from those that cannot is vital to effective planning and resource allocation for future preparedness plans. To address this challenge, we have developed a simple method for refining uncertain risk assessments with readily available data on disease importations.

Using previously validated ecological suitability estimates as a starting point, we applied the method to update ZIKV  $R_0$  estimates for each of the 254 counties in Texas, and found that only six counties have non-negligible probabilities of sustained local transmission, though an additional 77 counties may also be at low risk if our assumptions about local case data are too liberal. This is a substantial downgrade in expected risk, given that 43% of the 254 counties were previously thought to be vulnerable to ZIKV outbreaks, and even counties still at risk for epidemics on average had  $R_0$  estimates downgraded by 80% [4]. Based on these estimates and the number of importations that occurred in the state, our refined model suggests that there should have been between zero and one detected case of locally acquired ZIKV between January and September of 2017, which corresponds to the single transmission event actually detected in Cameron County in July 2017. Interestingly, a model that incorporates the possibility of many more missed imported cases into the state of Texas fits the data better, with an expected number of cases close to one (Fig 4).

There is a discrepancy between our model predictions and the ZIKV case data: our final model predicts the highest local transmission risk to be in the summer across the state, but most cases of local ZIKV transmission occurred in the winter (6 out of 7). While these case numbers are low and may simply reflect random variation, the most recent dengue outbreak in South Texas peaked in November, concurrent with a large dengue epidemic in Northern Mexico, suggesting that fall may be a higher risk season for arbovirus transmission than our final predictions indicate [40]. The discrepancy likely stem from errors in our initial risk estimates, which serve as informative priors for our analysis. Our model does not incorporate a time lag between case importation, local transmission, and case detection, which has been previously estimated around two months [23, 36, 37]. Incorporating a two month lag in our model slightly increased the fit of the resulting predictions (BIC = 2.9), but does not produce qualitatively different results in the magnitude of county risk other than the temporal shift \$14 Fig. While these results were promising, we chose to present our baseline results instead of those from the lagged model, because the baseline model has been previously validated [6]. Importantly our simplistic lagging procedure does not properly encapsulate the mechanistic intricacies determining the relationship between environmental suitability, mosquito population growth, case importation, local transmission, and subsequent detection that account for the lags outlined in [36]. Our results support these time lags as a crucial element for predicting ZIKV risk, and highlight the need for future research into the mechanistic underpinnings to reduce bias and improve model fit and predictive accuracy even further.

The factors driving county transmission risk were primarily temperature and mosquito population numbers (occurrence probability). Remaining uncertainty in county risk stems from a lack of detected imported cases, as imported cases primarily helped to downgrade Zika risk in the absence of local transmission. This provides the counterintuitive explanation for why our model suggests low epidemic risk for Harris county (containing the city of Houston) even though it was by far the county with the most imported cases. Given comparable importation and local case data, this approach could readily update ZIKV transmission risk estimates for any county in the continental US.

Our estimation method relies on several simplifying assumptions, which we divide into those concerning this ZIKV test case of the updating procedure, and those concerning the updating procedure itself. For this specific implementation, we assumed that the shape of the secondary case distribution resembles that of dengue. Although we have no evidence to the contrary, this should be updated as ZIKV-specific estimates become available [35]. We also assumed that transmission is equally likely from imported and locally acquired cases. Imported cases may be less infectious than locally acquired cases for two reasons, leading us to underestimate local transmission risks. First, they may be more likely to receive care that limits transmission, although most ZIKV cases are inapparent or mild and do not require medical care [16], and second their local infectious periods may be shorter than those of autochthonous cases. This assumption could be relaxed in cases where data are available on the relative detection rates. In this case, we also do not consider the possibility of sexual transmission of ZIKV. While sexual transmission has occurred and may be important for specific populations [41], we assumed that mosquito-borne transmission is the dominant mode of infection. Our current

iteration of the model makes several simplifying assumptions related to how we handle the temporal dynamics of both the *a priori* risk estimation and the updating procedure. Our baseline transmission risk model does not take any outside events that could modulate future risk into consideration (e.g. rainfall in the previous month causing upticks in mosquito populations), and future model development will be necessary to account for these important temporal dynamics [23, 36, 37]. For example, approaches that account for variability in the serial interval between successive cases and distributed lagged effects of environmental variables on transmission potential may hold promise for addressing this limitation [42].

The updating method itself also requires several simplifying assumptions. First, it relies on informative Bayesian priors for the relative spatiotemporal risk of a region. Posterior estimates could be biased by errors in these priors. At this point, we lack sufficient data from Texas to improve our estimates of the component parameters of a priori  $R_0$ . The lack of such data, in fact, motivated our more modest goal of refining estimates of relative risk across counties. We could potentially improve model accuracy by incorporating smoothing techniques that more realistically capture spatiotemporal correlations and heterogeneity in transmission risk across Texas. This method also treats all importations as independent. However, spatiotemporal heterogeneity in case detection probabilities or clustering of cases (e.g., testing of travel companions) could bias risk estimates. When secondary clusters are detected, we assume they share a transmission tree stemming from a single detected importation. In this case, the low ZIKV detection rate suggests that both primary importations and secondary cases are likely to be missed. If the detection rates between the two types of cases are roughly similar our results hold. If importations are detected at higher rates than secondary cases, then the resulting risk estimates will be higher; when we assume the reverse, meaning locally transmitted cases are more likely to be detected than imported cases, the posterior risk estimates will be reduced. The additional assumption, that clusters are epidemiologically connected, seems reasonable for the small contained outbreaks detected in Texas, but may not be appropriate for importation-fueled arbovirus outbreaks in Florida, for example. In such cases, molecular data might enable estimation of transmission clusters [43, 44]. Furthermore, there are identifiability issues associated with estimating the statewide scaling factor alongside the case detection probability, as both parameters modulate the expected number of secondary cases in the same fashion. We therefore used a strong, informative prior for the case detection probability from a previous study, and found no changes in the posterior distribution S5 Fig. If case detection is lower than we assumed, then the detected cases in Texas would have had to arise from larger outbreaks, so our final risk estimates would be closer to the prior (larger). Alternatively, if case detection in Texas is higher than we assumed, there would be an even larger discrepancy between our prior and resultant posterior risk estimates S15 Fig.

During the height of the ZIKV threat, public health agencies in the US rapidly implemented both preventative measures (e.g., vector control and educational campaigns) and response measures (e.g., laboratory testing and epidemic trigger plans), particularly in highrisk southern states. Decision makers sought to identify and narrow the spatiotemporal scope of outbreak risk to enable targeted responses, efficiently allocate resources, and avoid false alarms [15, 45]. Our method facilitates such rapid, real-time geographic risk estimation from typical early outbreak data, and allows for real-time updating of estimates as new data arise. In interpreting our results from 2016 in retrospect, they would suggest that the baseline transmission risk combined with the public health response was sufficient to mitigate the threat of a ZIKV epidemic across the state. Our method cannot disentangle the impact of public health response from the underlying transmission risk, so we cannot estimate the impact of the public health interventions. In real time, our results would have validated the magnitude of the public health response through the summer of 2016, and likely alleviated some of the epidemic concerns that arose in the winter when a handful of locally transmitted cases were detected in the state.

Critically, we can conclude neither that all initial ecological risk assessments for ZIKV will overestimate risk, although this seems to be the case for ZIKV in Texas and elsewhere in the US [9], nor that public health preparations and interventions for ZIKV are no longer necessary in Texas or the southern US. Rather, our results suggest that sustained ZIKV outbreaks are unlikely, but not impossible (3% of the state remains at low epidemic risk), and provide more robust and localized estimates of ZIKV risk that can inform more targeted surveillance and reactions to future ZIKV importations. While our method provides an avenue to improve imperfect *a priori* risk estimates using real-time data, future predictions and estimates could be improved with advances to our mechanistic understanding of transmission risk (e.g., incorporating rainfall or improved understanding of mosquito-human contact patterns) [2]. These estimates should also only be taken as the baseline level of risk for the state. Climate change, anomalous weather events, or unforeseen circumstances that alter mosquito populations or mosquito-human interactions could raise or lower the Zika risk in Texas [3, 46].

This framework is novel in its integration of *a priori* ecological transmission risk estimates with updating directly from real-time case reports [13, 14]. It thereby provides increasingly accurate and precise risk assessments to support public health decision making, and can be generalized to update  $R_0$  estimates from importation data, regardless of the *a priori* method of estimation. For example, a new approach combining epidemiological and molecular analyses suggests that transmission risk in Florida is subcritical (i.e.,  $R_0 < 1$ ) [44, 47]. Given that Florida experienced thousands of introductions, only a few of which sparked large outbreaks, coupling such outbreak-driven estimation with our terminal importation method may provide a powerful real-time risk assessment framework for exploiting all available data. This method resembles those used to assess disease transmission risk during elimination efforts, including malaria in non-endemic regions [48]. The key innovation is that, by starting with ecological suitability maps, we simultaneously identify important transmission hotspots and leverage case data from one region to inform risk estimates elsewhere.

We presented a simple and rational method for continuously updating transmission risk estimates for populations experiencing infectious disease importations, with or without secondary transmission. As we demonstrated for ZIKV in Texas, large numbers of terminal importations can profoundly lower both estimated risks of transmission and uncertainty in prior estimates, particularly those derived from ecological suitability or other models that borrow inputs from related pathogens in other parts of the world. Expanding our model to other regions could sharpen transmission risk estimates, and allow for more targeted public health interventions in the remaining hot spot locations [5, 9, 17]. Transmission risk estimates always include uncertainty, but by assimilating data in real-time, our method can help confirm or revise public health understanding in the midst of an outbreak. Although the threat of ZIKV emergence in the continental US motivated this study, this new framework can be applied to improve transmission risk assessments when a disease newly threatens a population via regular introductions with minimal secondary transmission.

#### Supporting information

**S1 File. Supplemental information.** (PDF)

S1 Fig. Sensitivity analysis of posterior  $R_0$  estimates for each county. Each point indicates a county-month posterior  $R_0$  estimate under different estimation scenarios. The x-axis value for all points is determined by the baseline scenario where posterior  $R_0$  estimates consider only a

single case detected in the November. The y-axis value is based on three sensitivity scenarios: (1) posterior estimates assuming no secondary transmission (dark grey), (2) five cases of secondary transmission in November (black), or (3) a single case of secondary transmission, but increased overall importations (light grey). Points falling above the black dashed line indicate that that a given scenario increases posterior  $R_0$  estimates compared to baseline estimates, and points below the line indicate the opposite. Estimates are compared for the median (left), and the 99th percentile (right) of the county-month distributions. Posterior  $R_0$  estimates increase if more secondary transmission is assumed, and decrease if less secondary transmission occurs, or if the absolute number of importations is increased.

(TIFF)

S2 Fig. Comparison between the empirical prior transmission risk distributions (points) and the fitted gamma distribution for three example county-month combinations (lines). (TIFF)

S3 Fig. Comparison between analytical likelihood predictions and simulations. We compare the probability mass functions for the outbreak sizes for our simulations (bars) with the analytical expectation (red dots). Rows demonstrate four different transmission risk scenarios ( $R_0$ ), and columns describe three different scenarios: (1) where every case within a transmission chain is detected (Perfect), (2) where all cases are detected independently with a specific reporting rate (Imperfect), and (3) where all cases are detected independently with a specific reporting rate except for the index case which is always detected (Imperfect Import). The Imperfect Import probability mass function is the one used for all analyses in this article. All simulations are completed with a reporting probability of 0.0574, and k = 0.12. (TIFF)

**S4 Fig. Match between estimated and assumed dispersion parameter.** Probability of a single importation generating 20 secondary infections from (29) (Line), or using our assumed dispersion parameter and negative binomial distribution (Points). (TIFF)

S5 Fig. Comparison between the informative reporting rate prior distribution (Red line) and the posterior distribution (histogram). (TIFF)

**S6 Fig. Trace plots used for assessing MCMC convergence.** The plots detail the posterior distribution for alpha assuming the actual temperature from 2016 and 2017, and either 0, 1, or 5 detected cases in the year.

(TIFF)

**S7 Fig.**  $R_0$  updating using importation data. Consider a hypothetical scenario in which the first 15 terminal ZIKV importations into Texas arrive in Harris county (which includes Houston) during August 2016. (A) Estimated Harris county  $R_0$  for August 2016 *a priori* (dark grey) and after accounting for the 15 (light grey) terminal importations (Future August). These distributions are composed of 1,000 samples from the prior and posterior distributions (respectively). (B) Median  $R_0$  estimates for August before (August 2016) and following (Future August) the importation-based update. (TIFF)

**S8 Fig. Prior**  $R_0$  estimates for Texas counties for each month. Median and 99 percentiles are shown for each County. Fill color indicates the estimated Median or 99 percentile estimate for that county for the given month, with counties showing yellow or red indicating their  $R_0$  is

above one (labels). Estimates are made for each month based on the average monthly temperature for 2016.

(TIFF)

S9 Fig. Comparison between prior  $R_0$  estimates using the historic average temperature for months versus using the actual temperature from the months in 2016. (TIFF)

S10 Fig. Posterior  $R_0$  estimates for Texas counties for each month using the actual 2016 temperatures. Fill color indicates the estimated Median or 99 percentile estimate for that county for the given month, with counties showing yellow or red indicating their  $R_0$  is above one (labels). Estimates are made using all importations through December of 2016, and assuming a single transmission event in both November and December. (TIFF)

S11 Fig. Posterior  $R_0$  estimates for Texas counties for each month using historic temperature data for running estimation procedure and priors. Fill color indicates the estimated Median or 99 percentile estimate for that county for the given month, with counties showing yellow or red indicating their  $R_0$  is above one (labels). Estimates are made using all importations through December of 2016, and assuming a single transmission event in November. (TIFF)

S12 Fig. Comparison between the prior and posterior  $R_0$  estimates as a function of the month of the year. Each point corresponds to a specific county in the state, and the colors indicate the number of importations that the county experiences during the specific month. (TIFF)

S13 Fig. Comparison between the posterior distributions of  $\alpha$  from the baseline (Normal) scenario with the lagged (Lagged) scenario. (TIFF)

S14 Fig. Posterior median county  $R_0$  estimates for Texas, based on ZIKV importations through January 2017. This assumes that all importations were terminal except for two autochthonous cases detected in Cameron County in late 2016, and shifts the prior transmission risk estimates by two months compared with the baseline scenario. (TIFF)

**S15 Fig. Posterior estimate comparison based on changes to the the case detection probability.** Dashed line indicates what the posterior estimates would be like if they matched the baseline posterior risk estimates. Colored points identify posterior estimates for different case detection probabilities. Yellow points are for a scenario where the probability for detecting cases was halved, and the purple points indicate estimates for when the probability was doubled.

(TIFF)

#### Acknowledgments

The authors would like to acknowledge Texas DSHS for providing the county-level ZIKV importation data used in this analysis. We would also like to acknowledge the Texas Advanced Computing Center (TACC) at The University of Texas at Austin for providing high performance computing resources that have contributed to the research results reported within this paper. URL: http://www.tacc.utexas.edu.

#### **Author Contributions**

Conceptualization: Spencer J. Fox, Steven E. Bellan, Michael A. Johansson.

Data curation: Spencer J. Fox.

Formal analysis: Spencer J. Fox, Steven E. Bellan.

Funding acquisition: Lauren Ancel Meyers.

Investigation: Spencer J. Fox.

Methodology: Spencer J. Fox, T. Alex Perkins, Michael A. Johansson, Lauren Ancel Meyers.

Resources: Spencer J. Fox, Lauren Ancel Meyers.

Software: Spencer J. Fox, Steven E. Bellan, Lauren Ancel Meyers.

Supervision: Steven E. Bellan, Lauren Ancel Meyers.

Validation: Spencer J. Fox.

Visualization: Spencer J. Fox.

Writing - original draft: Spencer J. Fox.

Writing – review & editing: Spencer J. Fox, Steven E. Bellan, T. Alex Perkins, Michael A. Johansson, Lauren Ancel Meyers.

#### References

- Grills A, Morrison S, Nelson B, Miniota J, Watts A, Cetron MS. Projected Zika Virus Importation and Subsequent Ongoing Transmission after Travel to the 2016 Olympic and Paralympic Games—Country-Specific Assessment, July 2016. MMWR Morb Mortal Wkly Rep. 2016; 65(28):711–715. https://doi.org/ 10.15585/mmwr.mm6528e1 PMID: 27442184
- Ali S, Gugliemini O, Harber S, Harrison A, Houle L, Ivory J, et al. Environmental and Social Change Drive the Explosive Emergence of Zika Virus in the Americas. PLOS Neglected Tropical Diseases. 2017; 11(2):1–16. https://doi.org/10.1371/journal.pntd.0005135
- Reiter P, Lathrop S, Bunning M, Biggerstaff B, Singer D, Tiwari T, et al. Texas lifestyle limits transmission of dengue virus. Emerg Infect Dis. 2003; 9(1):86–89. <u>https://doi.org/10.3201/eid0901.020220</u> PMID: 12533286
- Castro LA, Fox SJ, Chen X, Liu K, Bellan SE, Dimitrov NB, et al. Assessing real-time Zika risk in the United States. BMC Infect Dis. 2017; 17(1):284. <u>https://doi.org/10.1186/s12879-017-2394-9</u> PMID: 28468671
- Carlson CJ, Dougherty ER, Getz W. An Ecological Assessment of the Pandemic Threat of Zika Virus. PLoS Negl Trop Dis. 2016; 10(8):e0004968. https://doi.org/10.1371/journal.pntd.0004968 PMID: 27564232
- Alex Perkins T, Siraj AS, Ruktanonchai CW, Kraemer MUG, Tatem AJ. Model-based projections of Zika virus infections in childbearing women in the Americas. Nature Microbiology. 2016; 1(9):16126. https://doi.org/10.1038/nmicrobiol.2016.126 PMID: 27562260
- Lloyd-Smith JO, Schreiber SJ, Kopp PE, Getz WM. Superspreading and the effect of individual variation on disease emergence. Nature. 2005; 438(7066):355–359. https://doi.org/10.1038/nature04153 PMID: 16292310
- Wallinga J, Lipsitch M. How generation intervals shape the relationship between growth rates and reproductive numbers. Proc Biol Sci. 2007; 274(1609):599–604. https://doi.org/10.1098/rspb.2006.3754
   PMID: 17476782
- Carlson CJ, Dougherty E, Boots M, Getz W, Ryan SJ. Consensus and conflict among ecological forecasts of Zika virus outbreaks in the United States. Sci Rep. 2018; 8(1):4921. https://doi.org/10.1038/ s41598-018-22989-0 PMID: 29563545
- Hernandez PA, Graham CH, Master LL, Albert DL. The effect of sample size and species characteristics on performance of different species distribution modeling methods. Ecography. 2006; 29(5):773– 785. https://doi.org/10.1111/j.0906-7590.2006.04700.x

- Escobar LE, Craft ME. Advances and Limitations of Disease Biogeography Using Ecological Niche Modeling. Front Microbiol. 2016; 7:1174. https://doi.org/10.3389/fmicb.2016.01174 PMID: 27547199
- Heffernan JM, Smith RJ, Wahl LM. Perspectives on the basic reproductive ratio. J R Soc Interface. 2005; 2(4):281–293. https://doi.org/10.1098/rsif.2005.0042 PMID: 16849186
- Blumberg S, Lloyd-Smith JO. Inference of R(0) and transmission heterogeneity from the size distribution of stuttering chains. PLoS Comput Biol. 2013; 9(5):e1002993. <u>https://doi.org/10.1371/journal.pcbi.</u> 1002993 PMID: 23658504
- Blumberg S, Lloyd-Smith JO. Comparing methods for estimating R0 from the size distribution of subcritical transmission chains. Epidemics. 2013; 5(3):131–145. <a href="https://doi.org/10.1016/j.epidem.2013.05.002">https://doi.org/10.1016/j.epidem.2013.05.002</a> PMID: 24021520
- Hahn MB, Eisen RJ, Eisen L, Boegler KA, Moore CG, McAllister J, et al. Reported Distribution of Aedes (Stegomyia) aegypti and Aedes (Stegomyia) albopictus in the United States, 1995-2016 (Diptera: Culicidae). J Med Entomol. 2016;. https://doi.org/10.1093/jme/tjw072
- Duffy MR, Chen TH, Thane Hancock W, Powers AM, Kool JL, Lanciotti RS, et al. Zika Virus Outbreak on Yap Island, Federated States of Micronesia. N Engl J Med. 2009; 360(24):2536–2543. https://doi. org/10.1056/NEJMoa0805715 PMID: 19516034
- Messina JP, Kraemer MU, Brady OJ, Pigott DM, Shearer FM, Weiss DJ, et al. Mapping global environmental suitability for Zika virus. Elife. 2016; 5. https://doi.org/10.7554/eLife.15272 PMID: 27090089
- Nordhaus WD. Geography and macroeconomics: new data and new findings. Proc Natl Acad Sci U S A. 2006; 103(10):3510–3517. https://doi.org/10.1073/pnas.0509842103 PMID: 16473945
- Hijmans RJ, Cameron SE, Parra JL, Jones PG, Jarvis A. Very high resolution interpolated climate surfaces for global land areas. Int J Climatol. 2005; 25(15):1965–1978. https://doi.org/10.1002/joc.1276
- 20. NOAA. PSD Gridded Climate Datasets; 2017. Available from: https://www.esrl.noaa.gov/psd/.
- Kraemer MUG, Sinka ME, Duda KA, Mylne AQN, Shearer FM, Barker CM, et al. The global distribution of the arbovirus vectors Aedes aegypti and Ae. albopictus. Elife. 2015; 4:e08347. https://doi.org/10. 7554/eLife.08347 PMID: 26126267
- 22. Texas Department of State Health and Human Services. Zika in Texas; 2017.
- Mordecai EA, Cohen JM, Evans MV, Gudapati P, Johnson LR, Miazgowicz K, et al. Detecting the impact of temperature on transmission of Zika, dengue and chikungunya using mechanistic models. PLOS Neglected Tropical Diseases. 2017; 11:e0005568. https://doi.org/10.1371/journal.pntd.0005568 PMID: 28448507
- 24. Tesla B, Demakovsky LR, Mordecai EA, Bonds MH, Ngonghala CN, Brindley MA, et al. Temperature drives Zika virus transmission: evidence from empirical and mathematical models. Proceedings of the Royal Society B. 2018; 285:20180795. https://doi.org/10.1098/rspb.2018.0795 PMID: 30111605
- Brady OJ, Johansson MA, Guerra CA, Bhatt S, Golding N, Pigott DM, et al. Modelling adult Aedes aegypti and Aedes albopictus survival at different temperatures in laboratory and field settings. Parasit Vectors. 2013; 6(1):351. https://doi.org/10.1186/1756-3305-6-351 PMID: 24330720
- 26. Perkins A, Siraj A, Warren Ruktanonchai C, Kraemer M, Tatem A. Model-based projections of Zika virus infections in childbearing women in the Americas. bioRxiv. 2016.
- Nishiura H, Halstead SB. Natural history of dengue virus (DENV)-1 and DENV-4 infections: reanalysis of classic studies. J Infect Dis. 2007; 195(7):1007–1013. <u>https://doi.org/10.1086/511825</u> PMID: 17330791
- Scott TW, Morrison AC, Lorenz LH, Clark GG, Strickman D, Kittayapong P, et al. Longitudinal studies of Aedes aegypti (Diptera: Culicidae) in Thailand and Puerto Rico: population dynamics. J Med Entomol. 2000; 37(1):77–88. https://doi.org/10.1603/0022-2585-37.1.77 PMID: 15218910
- Reiner RC Jr, Perkins TA, Barker CM, Niu T, Chaves LF, Ellis AM, et al. A systematic review of mathematical models of mosquito-borne pathogen transmission: 1970-2010. J R Soc Interface. 2013; 10 (81):20120921. https://doi.org/10.1098/rsif.2012.0921 PMID: 23407571
- Muir LE, Kay BH. Aedes aegypti survival and dispersal estimated by mark-release-recapture in northern Australia. Am J Trop Med Hyg. 1998; 58(3):277–282. https://doi.org/10.4269/ajtmh.1998.58.277 PMID: 9546403
- Chan M, Johansson MA. The incubation periods of Dengue viruses. PLoS One. 2012; 7(11):e50972. https://doi.org/10.1371/journal.pone.0050972 PMID: 23226436
- Zhang Q, Sun K, Chinazzi M, Pastore Y Piontti A, Dean NE, Rojas DP, et al. Spread of Zika virus in the Americas. Proc Natl Acad Sci U S A. 2017; 114(22):E4334–E4343. <u>https://doi.org/10.1073/pnas.</u> 1620161114 PMID: 28442561
- 33. Tesla B, Demakovsky LR, Packiam HS, Mordecai EA, Rodríguez AD, Bonds MH, et al. Estimating the effects of variation in viremia on mosquito susceptibility, infectiousness, and R0 of Zika in Aedes

aegypti. PLOS Neglected Tropical Diseases. 2018; 12(8):1–19. https://doi.org/10.1371/journal.pntd. 0006733

- Riehle MM, Bukhari T, Gneme A, Guelbeogo WM, Coulibaly B, Fofana A, et al. The Anopheles gambiae 2La chromosome inversion is associated with susceptibility to *Plasmodium falciparum* in Africa. eLife. 2017; 6:e25813. https://doi.org/10.7554/eLife.25813 PMID: 28643631
- Padmanabha H, Durham D, Correa F, Diuk-Wasser M, Galvani A. The interactive roles of Aedes aegypti super-production and human density in dengue transmission. PLoS Negl Trop Dis. 2012; 6(8): e1799. https://doi.org/10.1371/journal.pntd.0001799 PMID: 22953017
- 36. Stewart-ibarra AM, Lowe R. Climate and Non-Climate Drivers of Dengue Epidemics in Southern Coastal Ecuador. Am J Trop Med Hyg. 2013; 88(5):971–981. https://doi.org/10.4269/ajtmh.12-0478 PMID: 23478584
- Siraj AS, Alex Perkins T. Assessing the population at risk of Zika virus in Asia—is the emergency really over? BMJ Global Health. 2017; 2(3):e000309. <u>https://doi.org/10.1136/bmjgh-2017-000309</u> PMID: 29082009
- Asher J, Barker C, Chen G, Cummings D, Chinazzi M, Daniel-Wayman S, et al. Preliminary results of models to predict areas in the Americas with increased likelihood of Zika virus transmission in 2017; 2017.
- Colón-González FJ, Peres CA, Steiner São Bernardo C, Hunter PR, Lake IR. After the epidemic: Zika virus projections for Latin America and the Caribbean. PLoS Negl Trop Dis. 2017; 11(11):e0006007. https://doi.org/10.1371/journal.pntd.0006007 PMID: 29091713
- 40. Thomas DL, Santiago GA, Abeyta R, Hinojosa S, Torres-Velasquez B, Adam JK, et al. Reemergence of Dengue in Southern Texas, 2013. Emerg Infect Dis. 2016; 22(6):1002–1007. <u>https://doi.org/10.3201/</u> eid2206.152000 PMID: 27191223
- Allard A, Althouse BM, Hébert-Dufresne L, Scarpino SV. The risk of sustained sexual transmission of Zika is underestimated. PLoS Pathog. 2017; 13(9):e1006633. https://doi.org/10.1371/journal.ppat. 1006633 PMID: 28934370
- 42. Oidtman RJ, Lai S, Huang Z, Siraj AS, Reiner RC, Tatem AJ, et al. Inter-annual variation in seasonal dengue epidemics driven by multiple interacting factors in Guangzhou, China. bioRxiv. 2018.
- **43.** Turk T, Bachmann N, Kadelka C, Böni J, Yerly S, Aubert V, et al. Assessing the danger of self-sustained HIV epidemics in heterosexuals by population based phylogenetic cluster analysis. Elife. 2017; 6. https://doi.org/10.7554/eLife.28721 PMID: 28895527
- 44. Grubaugh ND, Ladner JT, Kraemer MUG, Dudas G, Tan AL, Gangavarapu K, et al. Genomic epidemiology reveals multiple introductions of Zika virus into the United States. Nature. 2017; 546(7658):401– 405. https://doi.org/10.1038/nature22400 PMID: 28538723
- 45. Centers for Disease Control and Prevention. CDC Zika Interim Response Plan; 2016.
- Hotez PJ, Jackson Lee S. US Gulf Coast states: The rise of neglected tropical diseases in "flyover nation". PLOS Neglected Tropical Diseases. 2017; 11(11):1–4. https://doi.org/10.1371/journal.pntd. 0005744
- Dinh L, Chowell G, Mizumoto K, Nishiura H. Estimating the subcritical transmissibility of the Zika outbreak in the State of Florida, USA, 2016. Theor Biol Med Model. 2016; 13(1):20. https://doi.org/10. 1186/s12976-016-0046-1 PMID: 27829439
- Churcher TS, Cohen JM, Novotny J, Ntshalintshali N, Kunene S, Cauchemez S. Public health. Measuring the path toward malaria elimination. Science. 2014; 344(6189):1230–1232. <a href="https://doi.org/10.1126/science.1251449">https://doi.org/10.1126/science.1251449</a> PMID: 24926005